

JRC TECHNICAL REPORTS

From R&D to market: using trademarks to capture the market capability of top R&D investors

JRC Working Papers on Corporate R&D and Innovation No 01/2018

Carolina Castaldi, Mafini Dosso 2018



This publication is a Technical report by the Joint Research Centre (JRC), the European Commission's science and knowledge service. It aims to provide evidence-based scientific support to the European policy-making process. The scientific output expressed does not imply a policy position of the European Commission. Neither the European Commission nor any person acting on behalf of the Commission is responsible for the use which might be made of this publication.

Contact information Antonio Vezzani Address: Edificio Expo. c/ Inca Garcilaso, 3. E-41092 Seville (Spain) E-mail: <u>irc-b3-secretariat@ec.europa.eu</u> Tel.: +34 954488463 Fax: +34 954488316

JRC Science Hub https://ec.europa.eu/jrc

JRC110520 ISSN 1831-9408 (online)

Seville, Spain: European Commission, 2018

© European Union, 2018

Reproduction is authorised provided the source is acknowledged.

How to cite: Castaldi, C. and Dosso, M. (2018). From R&D to market: using trademarks to capture the market capability of top R&D investors, JRC Working Papers on Corporate R&D and Innovation No 01/2018, Joint Research Centre.

All images © European Union 2018

The **JRC Working Papers on Corporate R&D and Innovation** are published under the editorial supervision of Antonio Vezzani in collaboration with Andries Brandsma, Alex Coad, Fernando Hervás, Koen Jonkers, Pietro Moncada-Paternò-Castello, Alexander Tübke and Daniel Vertesy at the European Commission – Joint Research Centre; Michele Cincera (Solvay Brussels School of Economics and Management, Université Libre de Bruxelles); Enrico Santarelli (University of Bologna); Marco Vivarelli (Università Cattolica del Sacro Cuore, Milan).

The JRC Working Papers on Corporate R&D and Innovation addresses economic and policy issues related to industrial research and innovation and to the competitiveness of the European industry. Mainly addressed to policy analysts and the academic community, these are policy relevant early-stage scientific articles highlighting policy implications. These working papers are meant to communicate to a broad audience preliminary research findings, generate discussion and attract critical comments for further improvements. All papers have undergone a peer review process.

This Working Paper is issued in the context of the Industrial Research and Innovation Monitoring and Analysis (IRIMAII) activities that are jointly carried out by the European Commission's Joint Research Centre (JRC) – directorate B, Growth and Innovation and the Directorate General Research and Innovation - Directorate A, Policy Development and Coordination.

1

From R&D to market:

using trademarks to capture the market capability of top R&D investors¹

Carolina Castaldi

School of Innovation Sciences, Eindhoven University of Technology, The Netherlands <u>C.Castaldi@tue.nl</u>

Mafini Dosso

European Commission, Joint Research Centre, Seville, Spain <u>Mafini.dosso@ec.europa.eu</u>

Abstract

This paper investigates the links between the market capability of top corporate R&D investors (EU Industrial R&D Investment Scoreboards), as captured by trademark data and their economic performance in terms of net sales growth. It provides empirical evidence to better understand the extent to which companies, operating in different industrial sectors, combine technological capabilities with commercialization efforts to generate and appropriate the economic returns of their R&D investments. This paper shows how different dimensions of firms' market capabilities can be captured through trademark indicators. The results suggest that complementing R&D efforts and patenting activities with strong and specific market capabilities can indeed yield significant growth premiums. Moreover, offering services seems to pay off depending on the intensity of R&D investments. Yet, a quantile regression approach and a series of robustness checks indicate that such effects differ across the quantiles of the conditional sales growth distribution.

Keywords: R&D, trademarks, innovation, sales, services

JEL Classification: 032, 034, L10

¹ The authors would like to thank Alex Coad for his comments on previous drafted versions of the paper. The authors are also grateful to the organisers and reviewers of the Forum Innovation VII (Paris, France, 2016), the 14th ENEF Meeting on 'Technological change and employment dynamics at the firm and industry level'(2017, Pisa, Italy) and the CONCORDi 2017 10 years' anniversary conference (2017, Seville, Spain).

1. Introduction

Corporate R&D activities are highly concentrated, and the EU Industrial R&D Investment Scoreboards identify those key companies driving R&D efforts worldwide². The economic fates of these companies are strongly linked to their R&D budgets, as closely followed by market analysts. To have a real impact in their respective markets, these firms eventually have to turn their R&D into innovation output: new ideas, but also new products and services that successfully make it to the market and translate into increased sales revenues (Datta et al., 2015). Typical econometric studies investigating the impact of innovation on firm performance often include a direct link from R&D expenses to economic performance and assume an underlying mechanism of R&D translating into new ideas and then eventually new market offerings (Greenhalgh and Rogers, 2010). Yet, the road from R&D to market is a perilous one, even for the large companies with conspicuous R&D and marketing budgets. Companies' R&D efforts may fail to generate revenues from actual market sales, as firms differ in their capability to turn those inventions into actual new products and services sold in the marketplace. The latter capability entails very different skills from the ones relevant for R&D activities instead (Janssen et al., 2016). Eventually, innovation is a product of both technological and market capabilities of firms (Arora and Nandkumar, 2012). While patents have been used widely and meaningfully to gauge the inventive output of R&D investors and their technological capabilities in econometric studies, less attention has been given in these studies to the innovation output of R&D.

A recent empirical research strand has turned to trademarks as a way to capture downstream capabilities in general (Giarratana and Fosfuri, 2007; Arora and Nandkumar, 2012; Huang et al., 2013) and specifically the market introduction of innovation (Mendonça et al, 2004; Flikkema et al., 2014). Trademarks³ are distinctive signs (words, graphics, sounds, colours, etc.) associated with a good or service (WIPO, 2004). They fulfil two complementary roles. The first one is an identification role: they indicate the source or origin of a product. The second one is a differentiation role: they distinguish a good from competing offers by other entities in a given market (Ramello, 2006). A trademark owner has the exclusive right and also the obligation to use it in the market, the so-called 'use in market' requirement (Graham et al., 2013). While there is neither a novelty (as for patents) nor originality (as for copyrights) requirement to register a trademark, a new mark has to fulfil the main condition of 'distinctiveness' (WIPO, 2004). It must have a distinctive character from the viewpoint of the consumers, in such a way that it allows distinguishing and differentiating the products from the focal company among products in the market.

² For more information on the sample of companies included in the EU Industrial and R&D Investment Scoreboard, see <u>http://iri.jrc.ec.europa.eu/scoreboard.html</u>

³ Here we use the term trademarks to refer to both trade and service marks.

Trademark counts have been shown to be positively correlated to innovative capabilities at the firm level (Schautschick & Greenhalgh, 2016) and evidence at the trademark level also indicates that a large portion of trademarks does refer to innovation (Flikkema et al, 2014). Firstly, trademarks can capture innovation in later stages of the innovation process, namely commercialization (Nam and Barnett, 2011). A typical example is the standard strategy of pharmaceutical companies of protecting new chemical entities with patents and then building strong brands for the produced new drugs with trademarks and related marketing expenses (Reitzig, 2004). Here trademarks act as complements to R&D and patents. Secondly, trademarks can help identify non-patentable innovations, such as non-technological forms of innovation or service innovation (Flikkema et al, 2014). In this case, trademarks are really used as substitutes to patents.

Departing from this background, we aim at exploiting trademark data to capture the ability of top R&D investors to turn R&D into new market offerings. Our research question is: *to what extent does the market capability of top R&D investors, as captured by trademark data, explain their differential growth performance?*

Our analysis makes several original contributions to the innovation literature at the firm level. Firstly, we consider the combined effect of technological and market capabilities, capturing the latter with original indicators based on trademark data. Secondly, we investigate the effect of specific qualities of trademark portfolios, in particular their diversification and service intensity, thereby we extend recent studies that have only exploited trademark stocks. We also discuss how the indicators we propose have the potential to be used in the wider management research as well. Thirdly, we offer a new methodological approach by relying on quantile regressions, which allow us to gauge how the relation of our key variables to sales growth differs at different levels of the variables.

The remaining of the paper is organised as follows. Section 2 reviews the research background, focusing on recent studies exploiting trademark data. Section 3 details the methodological approach adopted for the study. Section 4 presents the results of the empirical analyses on the links between trademarks and economic performances. Section 5 concludes.

2. Research background

The economic rationale behind trademark systems rests upon the market failures that typically arise in case of information asymmetries between buyers and sellers (see the classical reference of Akerlof, 1970 and his market for 'lemons'). Trademarks can help overcome such market failures through two mechanisms (Economides, 1988). Firstly, they help reduce transaction costs by signalling the origin of the good, thereby reducing consumers' search costs. Secondly, they provide incentives for producers to invest in the quality of their products (Milgrom and Roberts, 1986). In this

interpretation, trademarks basically fulfil an informational function (Ramello and Silva, 2006).

Firms have several motives to register trademarks. These can be grouped into protection, marketing and exchange motives (Flikkema et al., 2014; Block et al., 2015, Castaldi, 2018). First, trademarks are used for their primary function of intellectual property rights, namely to protect from imitation (Hurmelinna-Laukkanen and Puumalainen, 2007). Second, they are used to support differentiation strategies (Ramello and Silva, 2006), which ultimately allow companies to earn excess returns thanks to the possibility of charging higher prices for branded products. In this sense, trademarks are the legal basis for building valuable brands and brand equity (Keller, 1993; Keller and Lehmann, 2006). Thirdly, trademarks can also be traded and used as leverage in external relations with suppliers, licensor, etc. (Graham et al., 2015).

2.1 Trademarks and innovation

A link between trademarks and innovation was convincingly put forward by Mendonça et al. (2004) a decade ago. Since then, several innovation researchers have picked up the challenge of developing innovation indicators based upon trademark data. Trademark-based indicators are now increasingly included in innovation rankings at the country level, such as the Innovation Union Scoreboard (EC, 2015) and policy reports (Dernis et al., 2015, OECD STI, 2015,).

Mendonça et al (2004) already discuss several reasons why trademark data could be used to construct meaningful indicators of innovation. Firstly, while patents refer more to inventions than innovations, trademarks can capture the commercialization of innovations. Trademarks are a tool for companies to signal new products in the market and marketing efforts can weigh as heavy as R&D efforts in the innovation value chain. This claim relies upon the informational role that trademarks fulfil. Whenever companies have invested in developing a new product, they will have the incentive to inform consumers in the marketplace of the quality of this innovation. Hipp and Grupp (2005) suggest indeed that trademarks may be a particular useful indicator for later stages of the innovation process, thereby providing complementary information to patent-based indicators (Llerena and Millot, 2013).

Secondly, trademarks are widely used across sectors (EPO-OHIM, 2013), and they can help to capture innovation activities in low-tech sectors as well. While patents always refer to technological innovation, trademarks can indicate non-technological innovation such as marketing innovation or organizational innovation (Millot, 2009), but also service innovation (Schmoch, 2003). In this perspective, trademarks can be considered as substitutes for patenting, since they offer an alternative formal tool available to protect non-patentable forms of innovation (Flikkema et al, 2014, 2015). Thirdly, trademarks can be collected systematically from trademark offices⁴ and are classified as protecting a good/service or a combination thereof following the Nice classification system (WIPO, 2013a), often complemented with a detailed description of the type of goods/services covered (Schmoch and Gauch, 2009). They also represent a meaningful indicator at different levels of aggregation, firm, sector or country, which makes them suitable for use also in aggregate indicators of innovation such as the EU Innovation Scoreboard.

When focusing on services, early evidence of the opportunities of using trademarks to proxy for innovation is presented in Schmoch (2003). His measure of innovation is taken from the Community Innovation Survey (CIS) and refers to the share of sales coming from new products and new services. He does not only look at simple correlations between innovation and trademarks, but also considers control variables including sector, firm size and measures of firm human capital, that might affect the firms' innovative success. He finds that the correlation between trademarks use and innovation is particularly significant in knowledge-intensive services, where patents are instead less related to innovation.

The above studies are illustrative but they are typically not based on multivariate testing (except for Schmoch, 2003). Instead, a handful of other studies (Allegrezza and Guard-Rauchs 1999, Millot, 2012, Götsch and Hipp, 2012) have engaged in developing testable models of the firm-level propensity to trademark. The question is then: are innovative firms more likely to use trademarks, after controlling for other factors?

Allegrezza and Guard-Rauchs (1999) provides an early attempt based upon a sample of Benelux firms. Their measure of trademark use is a binary variable indicating whether the firms have registered trademarks at all. While confirming the positive correlation between trademark use and R&D activities at the firm level, they introduce a number of interesting additional explanatory factors. They find that firms are more likely to use trademarks when they are large, when they gauge the risk of imitation by competitors to be high and when they are active in terms of exports. More recently, Millot (2012) shows theoretically and empirically that innovating firms have more economic incentives to trademark than imitating ones. Given that consumers have to recognize and value the innovation, her model predicts that the positive effect of trademarking are mostly there for product and marketing innovation and less so for process or organizational innovation. As far as the empirical results, she considers the use of patents and finds that the correlation of trademark use and product innovation only disappears in high-tech manufacturing after controlling for patent use. This suggests that in all sectors except high-tech manufacturing trademarks do bear additional informational content with respect to new products.

⁴ Since 1996, European firms can apply for the EU trademark (previously named Community Trademarks) at the EUIPO (previously OHIM) and an online database allows searching all registered or pending marks. The American office (USPTO) also offers an online search with a complete coverage going back to 1984.

Another empirical strategy to establish the link between trademarks and innovation is to perform trademark-level research and investigate whether individual trademarks refer to innovation or not. This is the strategy followed by Flikkema et al (2014) and Flikkema et al. (2015). A complementary strategy is to combine information on trademark registrations and new products announcements to also study to what extent individual product innovations are protected by a trademark. This labourintensive type of research can only be conducted for selected sectors and this is indeed what Malmberg (2005) did. He reports significant heterogeneity in trademark use across sectors, when looking at manufacturing only. He finds that trademarks are poorly linked to new products in the electromechanical and automotive industry, while a consistently high share of new products in the pharmaceutical industry are protected with trademark registration. His main recommendation is that trademarkbased indicators of innovation are only relevant for sectors with intensive use of trademarks and with products targeted at consumers and professional end-users.

2.2 Trademarks, market capabilities and economic performance

If one takes seriously the claims that trademarks capture innovation capabilities of firms, then a positive relation between trademarks and firm performance will be expected. In this case, the main implication is that trademarks signal product and service innovation, which in turn allows firms to reap economic benefits from sales of novel, improved offerings in the marketplace (Greenhalgh and Rogers, 2012). More generally, several studies have also used trademarks to proxy for different forms of intangible assets, including reputation and brand equity (WIPO 2013b). Brands easily account for half or more of the market value of the some of the large global enterprises such as Apple or Google (Corrado and Hulten, 2010) and brand equity is found as a key driver of economic performance across sectors (Vomberg et al., 2015). Trademarks can be seen as a weapon in the 'semiotic struggle' (Mendonça, 2014) where companies increasingly engage in persuading consumers of their value propositions.

Schautschick and Greenhalgh (2016) offer a comprehensive review of empirical studies of trademarks in the economic literature. They discuss three main approaches for studying the relation between trademarks and economic performance: (1) market value studies, (2) profitability studies (3) survival studies. Most existing studies are market value studies, therefore we will focus on those here.

The overall picture stemming from these studies is that trademarks are positively related to market value, albeit under some conditions. For instance, Greenhalgh and Rogers (2012) systematically investigate a large sample of UK firms and study both the effect of trademark activity as such (comparing companies owning at least a trademark and companies owning none) and of trademark intensity. While they do find a strong effect on market value for the trademark activity dummy, the results for trademark intensity are less clear cut and suggest in fact a non-linear relation, with decreasing returns appearing at higher levels of trademark intensity.

Sandner and Block (2011) look at the combined effects of R&D, patents and trademarks and find a strong effect of trademark stocks on firms' valuations. Information on trademark stocks partly overlaps with information on patent counts, suggesting complementarity of the two indicators. Measures of value of trademarks are also used. Interestingly, trademark breadth, as measured by the number of Nice classes covered by companies' trademark portfolios seem to imply a discount in market valuations for highly diversified companies. In more recent work, Block, Sandner and co-authors suggest going beyond merely counting trademarks and propose a classification of new trademark filings under different brand strategies, such as brand extending, modernizing and creating (Block et al, 2014). The empirical results show that market value is positively affected by trademark filings aimed at extending existing brands, while the effect is negative for creation of new brands. This might be inconsistent with the idea that new trademarks refer to innovation, but could also be taken as the starting point for further research on the linkages between trademark families and types of innovation, as Flikkema et al. (2015) attempt to do.

Dosso and Vezzani (2017) examine the impact of patents and trademarks on the financial markets' valuation of R&D-driven innovative firms. Their study is particularly relevant for our purposes here since they also investigate top R&D investors worldwide and accounts for the interactions between patents and trademarks. Their original contribution lies in distinguishing the effects of an increase in market value deriving from additional innovative output (labelled within-effects) with respect to the premium received for being more innovative than their competitors (labelled between-effects). In line with Zhou et al (2016)⁵, their study confirms the existence of patent-trademark complementarities in relation to the financial performance of top R&D investors, although specific industry patterns are observed.

In sum, the emerging empirical literature linking trademarks to economic performance provides clear evidence that trademarks entail positive returns to companies. The econometric studies rest, more or less explicitly, on a combination of theoretical mechanisms highlighted in the management and marketing literature, suggesting that trademarks capture reputational assets, innovation and downstream capabilities.

The main methodological limitations of this literature are two. First, most studies are market value studies, thereby they focus on intellectual property rights as intangible assets of companies and on valuation in financial markets. In our study we will shift attention to sales growth, to directly capture the impact of trademarks on changes in market sales. Second, the current studies have mostly used trademark stocks or even binary variables of trademark use, with only rare attempts towards extracting more information from trademark records. In our study, we develop novel firm-level

⁵ Zhou et al (2016) exploit a sample of start-ups to show the existence of complementary effects of patents and trademarks on the venture capital funding.

indicators able to shed light on the quality rather than the quantity of trademark portfolios.

3. Hypotheses development

Based upon the previous discussion, new trademarks can be used to capture the ability of Scoreboard companies to introduce new offerings in the market, what we will refer to as 'market capability'. We take stock of the ideas, reviewed in the previous section, that new trademark applications reveal process efforts (actual activities and commitments) related to the market introduction of innovation. In this interpretation, trademark portfolios of companies can proxy for underlying market capabilities. While some companies may be particularly skilled at developing technical inventions, they may have weaker capabilities when it comes to turning those inventions into new offers in the market. Company growth measured in net sales growth is well-suited to test for the effect of market capabilities. We propose the following:

Hypothesis 1: Market capability, as captured by trademark stocks, is positively related to top R&D investor's sales growth.

Yet, market capabilities can also be thought of as enhancing the effect of technological capabilities on sales growth. This is in relation with the interpretation that technological and market capabilities should be thought as complementary in innovation processes: they reinforce each other and refer to different phases of the innovation processes (Sandner and Block, 2011).

Hypothesis 2: For top R&D investors, market capability strengthens the relation between corporate technological capabilities and sales growth.

Firms can be compared in terms of their trademark counts, yet, an important caveat when working with trademark data is that trademark counts depend as much on firm-specific strategies as they depend on sectoral level specificities. Sectoral differences tend to be strongest in explaining differences in trademark strategy (Malmberg, 2005; Flikkema et al., 2014, Dosso and Vezzani, 2017), as there are sectors where it is common to trademark each new product model (like in the Automotive sector) and other sectors where companies will only trademark the company name (typically in low-tech service sectors). The idea that sectoral differences matter resonates with the broader debate within the innovation literature (Srholec and Verspagen, 2012). All in all, meaningful cross-firm comparisons are only possible within given sectors with similar tendencies to trademark.

Hypothesis 3: For top R&D investors, the relation of market capability with sales growth depends upon the industry considered.

While the size of their trademark portfolio may shed light on the strength of companies' market capabilities, the quality of the trademark portfolio also reveals salient information about the company's route to the market. We focus here on the diversification of the trademark portfolio. The top R&D investors are large companies with a significant level of product diversification. Related diversification, i.e. diversification that can rely to a large extent upon forms of economies of scale/scope, is typically seen as a driver of firm level economic performance (Teece, 1982; Teece et al, 1994). Empirical studies have looked at corporate diversification along different dimensions. One typical way is by considering the technological diversification of companies, as captured by the patent classes where firms own patents (Breschi et al 2003). Concentration in patent classes is the antagonist measure, indicating the extent to which companies focus their patenting efforts in a limited set of technological fields (Dernis et al, 2015). For technological diversification of the top R&D investors holds: first, average company diversification differs substantially across sectors; second, technological diversification is positively correlated to subsidiary diversification as measured by the number of industries covered by the companies' subsidiaries (Dernis et al, 2015, p. 34-35). An additional dimension of diversification can refer to the actual markets where companies are active, which can be traced by the market coverage of the trademarks owned by the companies. Measures of diversification using trademarks are rather novel, but there are seminal attempts in specific sectors (Gao and Hitt, 2012, Castaldi and Giarratana, 2018). There are two different ways in which we envision capturing trademark diversification:

- a. Diversification across Nice classes: trademarks are registered for specific market segments, classified in 45 Nice classes. Because of the 'use in market' requirement, Nice classes coverage is highly correlated to the actual markets where companies introduce their products. This study focuses on trademark portfolios breadth, indicating the number of Nice classes covered.
- b. Goods-service diversification: Nice classes cover both goods and services. Single trademarks can cover both goods and service classes. While goods classes are the traditionally preferred classes for R&D intensive firms, coverage of service classes indicates that the companies are also active in developing complementary services next to their new product development efforts. This study looks at the trademark portfolios service intensity.

As for trademark portfolio breadth, theories on corporate diversification suggest that diversification tends to have a curvilinear relation to economic performance (Palich et al., 2000). Diversification is positively related until a threshold after which diseconomies of scale/scope emerge. There is evidence for this mechanism when it comes to technological diversification, but testing this hypothesis using trademark diversification has not happened yet.

Hypothesis 4: There is an inverted-U relation between corporate trademark diversification and top R&D investors' sales growth.

As for trademark portfolios service intensity, services are increasingly seen as a source of additional profits for manufacturing firms: while profit margins on manufactured goods tend to be small due to strong competition and high product turnover, margins on services are higher (Vandermerwe and Rada, 1989; Cusumano et al., 2015). Manufacturing firms able to develop business models where services are involved are found to enjoy higher revenues and higher customer satisfaction (Janssen et al., 2016, Janssen and den Hertog, 2016). This study exploits the trademark portfolios service intensity as a measure of the market capability of offering services.

Hypothesis 5: There is a positive relation between the service intensity of trademark portfolios and sales growth for the top R&D investors.

4. Methods and data

4.1 Data

The starting point for this study is the list of Top R&D investors worldwide from the 2013 edition of the *EU Industrial R&D Investment Scoreboard* (European Commission, 2013). This data source includes historical economic and financial information on the world's largest R&D investing companies, accounting for more than 80% of business R&D expenditures worldwide. Data are collected from the companies' annual reports and accounts from Bureau van Dijk Electronic Publishing GmbH. The main variable is the R&D investment which corresponds to the cash investment funded by companies themselves. In addition to R&D figures, data such as net sales, operating profits and market capitalisation are provided. Companies are assigned to the country of registered office and refer to the ultimate parent company. Consistently, the industrial affiliation of parent companies is reported according to the 3-digit level of the Industry Classification Benchmark (ICB).⁶

In collaboration with the OECD, the European Commission-Joint Research Centre has collected patents and trademarks of the Scoreboard (SB) companies (about 2000 companies), edition 2013, relying on the companies and their structure of subsidiaries (more than 500. 000 subsidiaries) as reported in the ORBIS database.⁷ The report presenting the dataset (Dernis et al, 2015) offers a comprehensive view of the intellectual property rights (IPR) bundles of top R&D investors. As that report is essentially descriptive in nature, this study complements it by offering a first explanatory analysis taking stock of their matching efforts.

⁶ More details about the ICB classification can be found at <u>http://www.ftse.com/products/indices/icb</u>

⁷ ORBIS is a commercial database of Bureau van Dijk, which provides economic and financial company information.

In particular, we exploit here the matched data on all new trademarks applications filed between 2005 and 2012 at the USPTO⁸. This office is the preferred one for the trademark applications of most companies, except for Japan. This holds true also for their patenting activities (Dernis et al, 2015). We consider only new trademarks filed by the top R&D investors in the years considered, not all existing 'live' ones. This choice allows us focusing both on innovation-related activities and on current capabilities.

Trademarks are classified according to the Nice classification of goods and services, covering 45 classes: classes from 1 to 34 cover goods and classes 35-45 cover services (WIPO, 2013a). Each trademark can be applied for use in several markets, thus it can cover multiple classes. There are two reasons why companies do not file trademarks covering all classes. First, they have to show that the trademarks will actually be used in those markets. Secondly, fees typically increase with the number of trademarks.⁹

Dernis et al (2015) warn about a caveat when using the IPR data. The matching of patent and trademark data to top R&D investors has been done by assuming the 2012 corporate structure for all years. The authors suggest that this assumption might be easily fulfilled for the years 2010-2012, but for earlier years the matching of company and controlled subsidiaries to IPR data is likely to be less reliable. Because of this, this study will focus on explaining growth in the years 2010-2012 with trademark-based variables in 2010. Earlier years will nonetheless be used for the calculation of stock variables.

For the top R&D investors worldwide as listed in SB 2013 edition, the JRC has several company data available. This study will focus on the information on the main sector of activity, the year net sales figures and the R&D investment and patents all available for the years 2005-2012. Finally, it should be noted that trademarks are salient data for top R&D investors since these firms are all active in the registration of intellectual property rights.

4.2 Econometric strategy

The aim of the econometric analysis is to find empirical evidence supporting or rejecting the theoretical hypotheses on market-related antecedents of growth. We

⁸ The USPTO and EUIPO are the two main offices for the trademark application of top R&D investors. Given the different institutional requirements and market-specific corporate strategies, exploiting information from the two offices make sense, assuming that the related variables should bear explanatory power. However, when including patents and trademarks from the two offices, the EUIPO variables turned out to be non-significant, suggesting that their additional informational value is negligible. Therefore, we focus on USPTO applications only and include a US dummy to control for home-market bias.

⁹ Notice though that the European office has a system where the basic fee covers up to 3 Nice classes and proportional fees only apply from the fourth class.

follow the standard modelling approach of growth regressions where the Gibrat law of proportionate growth is taken as the baseline model. In Gibrat type of regressions, the logarithmic growth at time t is modelled as a function of the initial value of the relevant size. Greenhalgh et al. (2011) already applied this approach in an empirical study on the effect of trademarking and branding, on a sample of UK firms.

Firm growth rates, however measured, are well known to exhibit highly skewed distributions (Bottazzi and Secchi, 2006). This skewness makes the application of OLS estimates problematic since extreme values are misrepresented. One way to deal with a skewed distribution in the dependent variable is to exploit quantile regressions and rely on Least Absolute Deviation (LAD) estimation techniques to obtain better coefficient estimates (Coad and Rao, 2008). LAD estimates are based on minimizing the absolute deviation from the median instead of the squared deviation from the mean as in the case of Ordinary Least Squares (OLS) estimates. The median is a better reference for skewed distributions than the average and quantile regressions also allow checking how estimated coefficients vary across quantiles. After inspecting the growth distribution of Top R&D investors to establish non-normality, quantile regressions will be exploited. As noted by Yaffee (2002), standard errors of LAD estimates based on 100 replications. Unless otherwise noted all estimates reported in the results are obtained from bootstrapping.

4.3 Definition of variables

Dependent variable

In this study we focus on growth of net sales as the variable of theoretical interest. As already explained, this definition of growth fits well with the theoretical mechanisms behind the proposed hypothesis. In particular, firms active in applying trademarks are expected to be better able to introduce products and services in the market and this ability should translate directly in above average growth in sales.

The dependent variable will be the logarithmic growth of net sales measured in the last three years of data available.

GROWTH 2010-2012= ln(NET SALES 2012)- ln(NET SALES 2010)

Independent variables

Following the structure of Gibrat regressions, the first independent variable to be included is the initial value of net sales for the period considered, in logarithmic terms, $ln(NET SALES_{2010})$.

Another key independent variable included is the *R&D stock*, capturing investment in technological capabilities. Following the widely accepted definitions by Hall and Oriani (2006), we use a depreciation rate of 15%.

Variables of theoretical interest.

Market capability (H1, H2 and H3): the strength of market capability is operationalized by trademark stock in 2010. Trademark stocks are preferred to simple trademark counts since new trademarks are accumulated in trademark portfolios (Sandner and Block, 2011) and often more specifically in trademark families (Block et al., 2014; Flikkema et al. 2015). Given the availability of counts of new trademarks from 2005, the stocks rely on six years of data. In line with Greenhalgh et al. (2011), trademark stocks are calculated similarly to R&D stocks, with a depreciation rate of 15%. Depreciation of trademarks can be thought to happen for two main reasons. First, trademarks can be abandoned, either actively (by requesting so) or passively (by failing to pay renewal fees). A second reason relates to actual loss of value and relevance because of a new market focus. In this sense, trademarks bear similar properties to other knowledge assets, but the knowledge involved is of a different nature, namely market-related and symbolic (Ramello and Silva, 2006). According to Sandner and Block (2011), who opt instead for a null depreciation rate, trademark abandonment is a rare event. Yet, we consider the second mechanism, i.e. the loss of relevance for the company to be a much more likely process and we stick to the 15% depreciation rate.

TM stock 2010 = new TMs 2009+(1-depreciation rate)* TM stock 2009

Trademark diversification (H4 and H5): in this study we focus on two measures of diversification.

The first is the average breadth of trademark portfolios, as measured by the average number of Nice classes covered by the trademarks filed by the top R&D investors for each year. The actual variable included in the model is the 3-years average in the period 2008-2010, *BREADTH* ₂₀₀₈₋₂₀₁₀, to limit the effects of important annual variability. We refer to Haans et al. (2016) for a robust empirical test of curvilinear relations.

As for H5, we exploit the distinction of goods *vs* services classes in the Nice classification. For each trademark we can record whether it covers at least a Nice service class. Next, we can count the number of trademarks by company and by year that covers at least a service class. The service intensity of the trademark portfolio of a company in a given year is simply the share of trademarks covering at least one service class on the total trademarks filed. Notice that this is a rather lenient measure of service intensity and it includes also the cases of trademarks covering service classes only.

For all independent variables, sectoral means are included as well, to control for crosssectoral heterogeneity in R&D levels, trademarking, trademark breadth and service propensity.

5. Results

5.1 Descriptives

We start by investigating the statistical features of the distribution of net sales growth rates, the dependent variable of this study. As expected from previous literature, we find that this distribution is not Normal (see p-p plots comparing the distribution with a Normal one, Figure 1). Of course our sample is to a certain extent a relatively more homogenous set of companies than representative national samples of companies on which most industrial dynamics studies are based. Yet, the skewness of the growth rates distribution is confirmed, in this case mostly a negative skewness, possibly due the 2009 financial crisis.



Figure 1 – Normality Q-Q plot of the dependent variable, GROWTH 2010-2012

Figure 2 clearly shows that the distribution of trademark applications across companies is highly skewed, with very few companies filing a substantially higher number of trademarks. This will be taken into account in the analysis by including only logarithmic versions of the trademark counts and stocks variables. Moreover, trademarking is highly sector-specific (Dernis et al 2015); this calls for including sectoral averages of all trademark-based independent variables as controls in the regression models.

Figure 2 – Distribution of USPTO trademarks applications, 2005-10: total number of trademarks (top) and its log (bottom).



Notes: LogTM=ln (1+TM). **Source:** EC-JRC/OECD, IP bundle of top corporate R&D investors, internal database.

Table 1 shows sectoral differences in our measure of service intensity, as defined earlier. For manufacturing sectors the shares of trademarks covering service classes is typically not higher than 30%. For service sectors the shares are generally higher. Table 2 summarizes all the firm-level variables defined for use in the regression models and Table 3 provides the related descriptive statistics.

3-digit code	Sector	%
530	Oil & Gas Producers	26%
570	Oil Equipment, Services & Distribution	30%
580	Alternative Energy	26%
1350	Chemicals	13%
1730	Forestry & Paper	12%
1750	Industrial Metals & Mining	12%
1770	Mining	18%
2350	Construction & Materials	21%
2710	Aerospace & Defence	29%
2720	General Industrials	10%
2730	Electronic & Electrical Equipment	14%
2750	Industrial Engineering	18%
2770	Industrial Transportation	29%
2790	Support Services	51%
3350	Automobiles & Parts	14%
3530	Beverages	10%
3570	Food Producers	6%
3720	Household Goods & Home Construction	11%
3740	Leisure Goods	24%
3760	Personal Goods	5%
3780	Tobacco	1%
4530	Health Care Equipment & Services	20%
4570	Pharmaceuticals & Biotechnology	16%
5330	Food & Drug Retailers	30%
5370	General Retailers	42%
5550	Media	47%
5750	Travel & Leisure	29%
6530	Fixed Line Telecommunications	53%
6570	Mobile Telecommunications	51%
7530	Electricity	25%
7570	Gas, Water & Multi-utilities	28%
8350	Banks	46%
8530	Non-life Insurance	100%
8570	Life Insurance	90%
8630	Real Estate Investment & Services	61%
8770	Financial Services	60%
8980	Equity Investment Instruments	0%
8990	Non-equity Investment Instruments	0%
9530	Software & Computer Services	48%
9570	Technology Hardware & Equipment	22%

Table 1 – Average sectoral service intensity as measured by the number of USPTO trademarks covering at least one service class, 2008-10

Notes. Details of ICB classification are available at http://www.ftse.com/products/indices/icb

Variable	Definition
Ln Net sales growth	Ln net sales in 2012-log net sales in 2010
Ln Net sales 2010	Ln of net sales, year 2010
Ln R&D stock 2010	Ln of R&D stock in 2010, data from 2005
Ln Patent stock 2010 U	Ln of Patents stock in 2010, data from 2005, patents filed at USPTO (U)
Ln TM stock 2010 U	Ln of Trademark stock in 2010, data from 2005, trademarks filed at USPTO (U)
Breadth TM U	Average number of Nice classes covered by trademarks filed at USPTO, mean of years 2008-2010
Service intensity U 2008-10	Share of trademarks filed at USPTO in the period 2008-2010 that cover at least a service Nice class
Sectoral ln R&D 2010	Ln of average level of R&D investment by 3-digit sectoral code
Sectoral Patent stock 2010 U	Average level of Ln Patent stock 2010 U by 3-digit sectoral code
Sectoral TM stock 2010 U	Average level of Ln Trademark stock 2010 U by 3-digit sectoral code
Sectoral breadth TM U	Average level of Breadth TM U by 3-digit sectoral code
Sectoral service intensity U 2008-10	Average level of Service intensity U 2008-10 by 3-digit sectoral code

Table 2 – List of variables and their definition.

Table 3 – Descriptive statistics of the variables used in the regressions.

Variable	Obs	Mean	Std. Dev.	Min	Max
Ln Net sales growth	1960	.1445485	.52358	-8.346049	8.08804
Ln Net sales 2010	1966	14.2895	1.937299	3.513275	19.45742
Ln R&D stock 2010	1546	12.73066	1.359978	8.253407	17.29352
Ln Patent stock 2010 U	2000	3.839361	2.157964	0	10.14685
Ln TM stock 2010 U	2000	2.726652	1.724485	0	8.091161
Breadth TM U	2000	1.279988	1.232801	0	14.66667
Service intensity U 2008-10	2000	.2204529	.2757574	0	1
Sectoral ln R&D 2010	2000	5.335257	.5618382	3.365055	6.331671
Sectoral Patent stock 2010 U	2000	3.839361	.8475246	0	5.142047
Sectoral TM stock 2010 U	2000	2.726652	.5303333	.6151857	4.72566
Sectoral breadth TM U	2000	1.279988	.2950423	.3333333	1.928428
Sectoral service intensity U 2008-10	2000	.2204529	.1275155	0	1

Interestingly, sectoral trademark intensity, as measured by number of trademark applications divided by sales, seems to be positively correlated with sectoral R&D intensity. As reported in Table 4, pairwise correlation of the actual variables included in our model reveals that R&D stocks and TM stocks are significantly positively correlated but not very highly so.

	Ln Net sales growt h	Ln Net sales 2010	Ln R&D stock 2010	Ln Paten t stock 2010 U	Ln TM stock 2010 U	Breadt h TM U	Service intensit y U 2008-10	Sectora l ln R&D 2010	Sectora l Patent stock 2010 U	Sectora l TM stock 2010 U	Sectora l breadt h TM U	Sectoral service intensit y U 2008-10
Ln Net												
growth	1											
Ln Net												
sales	-											
2010 Ln P&D	.2338*	1										
stock	-											
2010	.1041*	.6522*	1									
Ln												
Patent	_		6800									
2010 U	- .0996*	.3907*	*	1								
Ln TM												
stock	0007	0.7.4*	.5132	.5771	4							
2010 U Breadth	0097	.3/61*	* 3084	3505	1							
TM U	0014	.2864*	*	*	.6384*	1						
Service												
intensit			0754									
y 0 2008-10	.0075	.0043	.0754 *	.0164	.2580*	.1899*	1					
Sectoral							-					
ln R&D		-	.1675	.0888	-							
2010 Sectoral	.0004	.1728*	*	*	.0649*	1025*	0048	1				
Patent												
stock	-		.0820	.3927								
2010 U	.0625*	.0096	*	*	.0661*	.0165	2324*	.2262*	1			
Sectoral TM												
stock				.0845								
2010 U	.0185	.0574*	0181	*	.3075*	.1525*	0071	2111*	.2150*	1		
Sectoral												
TM II	.0041	.2395*	0146	.0271	.1959*	2393*	.0516*	4283*	.0690*	.6371*	1	
Sectoral		.2070	10110		11707		10010			10071	-	
service												
intensit												
y 0 00- 10	.0805*	- .1377*	0376	213*	0051	.0288	.4287*	0113	5420*	0166	.1203*	1

Table 4 – Pairwise correlation table among the variables used in the regressions

Notes: * denotes significant at 5% level or better

5.2 Hypotheses testing

In order to test the hypotheses presented in section 3, we estimate several models in which we include step-wise the different sets of variables of theoretical interest.

Testing for market capabilities

Table 5 reports coefficients from LAD quantile regression estimates of the models allowing testing for Hypotheses 1, 2 and 3 about market capabilities.

Model 1 is a baseline model where only prior level of net sales and technological capabilities are included, next to sectoral averages. The prior firm R&D stock already appears to be negatively related to net sales growth, similarly to the sectoral level of R&D. Large R&D investments seem to weigh heavily for these firms when we consider their relation to growth.

When we introduce the firm-level trademark stocks (Model 2), we find instead a positive and significant relation to sales growth. R&D stocks show negative significant relations, at the firm and at the sector level. Firm-level achievements in developing market capabilities as measured by the accumulation of new trademarks appear to give the top R&D spenders a clear growth premium.

	(1)	(2)	(3)
VARIABLES	Model 1	Model 2	Model 3
Ln Net sales 2010	-0.00931	-0.0156*	-0.0156*
	(0.00956)	(0.00935)	(0.00855)
Ln R&D stock 2010	-0.0147	-0.0228***	-0.0385***
	(0.00927)	(0.00846)	(0.00835)
Ln TM stock 2010 U		0.0175***	-0.0313
		(0.00473)	(0.0242)
interaction R&D * TM stocks			0.00381**
			(0.00181)
Sectoral ln R&D 2010	-0.772***	-0.310***	-0.287***
	(0.157)	(0.0648)	(0.0606)
Sectoral TM stock 2010 U		0.265***	0.244***
		(0.0619)	(0.0579)
US dummy	0.0824***	0.0629***	0.0582***
	(0.0125)	(0.0143)	(0.0136)
Constant	4.591***	1.578***	1.701***
	(0.851)	(0.180)	(0.175)
Sector dummies	YES	YES	YES
Pseudo R ²	0.0836	0.0897	0.0906
Observations	1,533	1,533	1,533

Table 5 – LAD quantile regression estimates for models testing Hypotheses 1, 2 & 3 (median regression estimates reported).

Notes: Standard errors in brackets, *** p<0.01, ** p<0.05, * p<0.1

Model 3 allows testing Hypothesis 3: interestingly the interaction effect of trademark stocks with R&D stocks is significant, while the direct effect of trademark stocks become insignificant. This supports Hypothesis 3 and at the same time casts doubts on

Hypotheses 2. In this model market capabilities appear to be enhancing the effect technological capabilities, in particular decrease the negative effect of R&D stocks.

Testing for trademark diversification

Hypotheses 4 and 5 concern two distinct dimensions of firm diversification in trademarking.

Models 4 and 5 include measures of firm-level and sectoral-level breadth, accounting also for a curvilinear relation. None of these variables turns out to have a significant relation to net sales growth of top R&D investors. Model 6 includes our measure of service intensity and this turns out to be significantly related to net sales growth. This is mostly a sectoral effect, but even the firm level service intensity has a significant effect. In this final model, the effect of trademark stocks via the interaction with R&D is confirmed.

Table 6 – LAD quantile regression estimates for models testing Hypotheses 4 & 5 (median regression estimates reported).

VARIABLES	Model 4	Model 5	Model 6
Ln Net sales 2010	-0.0157*	-0.0156*	-0.0198**
	(0.00861)	(0.00911)	(0.00872)
Ln R&D stock 2010	-0.0380***	-0.0388***	-0.0394***
	(0.00795)	(0.00906)	(0.00908)
Ln TM stock 2010 U	-0.0323	-0.0322	-0.0446*
	(0.0266)	(0.0299)	(0.0243)
interaction R&D * TM stocks	0.00387*	0.00386*	0.00480***
	(0.00198)	(0.00211)	(0.00184)
Breadth TM U	-0.000479	0.00161	
	(0.00562)	(0.0136)	
Breadth TM U squared		-0.000304	
1		(0.00190)	
ln_sectoralRDlevel	-0.132***	-0.132***	-0.134***
-	(0.0229)	(0.0225)	(0.0209)
Sectoral TM stock 2010 U	-0.114***	-0.115***	0.0267**
	(0.0400)	(0.0415)	(0.0115)
Sectoral breadth TM U	0.534***	0.537***	()
	(0.141)	(0.145)	
US dummy	0.0596***	0.0586***	0.0560***
	(0.0150)	(0.0148)	(0.0143)
Service intensity U 2008-10	(0.0100)	(0.0 - 10)	0.0651***
			(0.0241)
Sectoral service intensity U 2008-10			0.426***
			(0.128)
Constant	1 128***	1 1 38***	1 304***
Constant	(0.134)	(0.126)	(0.123)
Sector dummies	YES	YES	YES
Sector aumines	120	120	1.10
Pseudo R2	0.0897	0.0898	0.0914
Observations	1,533	1,533	1,533

Using the very last model (Model 6) as our final model, we exploit the possibilities of quantile regressions and investigate how the key estimated coefficients change when other quantiles than the median are used. Figures 3 reveal two interesting insights.

Firstly, OLS estimates would substantially overestimate the coefficients for R&D and underestimate the coefficients for TM stocks and service intensity. This confirms the relevance of using quantile regression techniques.

Secondly, while the estimated coefficients for R&D stocks show a relatively stable pattern across different quantiles, the picture stemming from the three other plots is quite different. In particular, the direct effect of trademark stocks is clearly declining and appears significantly negative for the top firms in terms of growth. Instead the moderating effect of trademark stocks captured by the interaction variable is increasing in importance and is particularly relevant for the same high growth firms. This suggests that it is this subset of the top R&D investors which mostly benefit from the complementarity between technological and market capabilities, but hardly benefit from focusing on higher levels of trademarking as such.

As for service intensity, the growth premium we already detected for the median firm seems to be there only for the in-between quantiles and completely disappear for the extreme quantiles, both top and low growth firms.

Figure 3 – Estimated coefficients from LAD regressions for R&D and trademark stock variables evaluated at different quantiles, confidence intervals included.



Dashed line represents OLS estimates with corresponding confidence intervals (dotted lines).

JRC Working Papers on Corporate R&D and Innovation – No. 01/2018

5.3 Robustness checks

In the models estimated so far we have included R&D stocks as proxies for technological capabilities. R&D can be viewed as an input measure of technological innovation, while patents are instead more of an output measure. We also estimate models where we use patent stocks as proxies for technological capabilities. The key results remain essentially unchanged for the median regressions. The only difference seems to be how the effect of service intensity changes across quantile estimators (Figure 4). In particular, these estimates show that a service focus benefits a larger group of firms than it was the case in the previous plots.

VARIABLES	Model 1	Model 2	Model 3
Ly Naticalas 2010	0 0 0 0 0 * * *	0 0 2 1 7 * * *	0 0 / 1 2 * **
Ln Net sales 2010	-0.0280^{+++}	-0.031/	-0.0413
La Datant at al 2010 II	(0.00602)	(0.00/8/)	(0.00647)
Ln Patent stock 2010 U	-0.00393	-0.0115	-0.0231
	(0.003/3)	(0.00385)	(0.00648)
Ln IM Stock 2010 U		$0.01/5^{***}$	0.000278
		(0.00407)	(0.00754)
interaction Patent * TM stocks			0.00472**
			(0.00166)
Sectoral Patent stock 2010 U	-0.0952***	-0.0565***	-0.0537***
	(0.0174)	(0.0142)	(0.0133)
Sectoral TM stock 2010 U		0.133***	0.118***
		(0.0434)	(0.0456)
US dummy	0.0801***	0.0559***	0.0493***
-	(0.0103)	(0.0106)	(0.0120)
Constant	0.838***	0.416***	0.613***
	(0.0541)	(0.0849)	(0.0903)
Sector dummies	YES	YES	YES
Pseudo R2	0.0683	0.0728	0.0763
Observations	1.960	1.960	1.960

Table 7 – Using patent variables only: LAD quantile regression estimates for models testing Hypotheses 1, 2 and 3 (median regression estimates reported).

Notes: Standard errors in brackets, *** p<0.01, ** p<0.05, * p<0.1

VARIABLES	Model 4	Model 5	Model 6
Ln Net sales 2010	-0.0417***	-0.0416***	-0.0402***
	(0.00612)	(0.00584)	(0.00567)
Ln Patent stock 2010 U	-0.0239***	-0.0239***	-0.0236***
	(0.00630)	(0.00595)	(0.00867)
Ln TM stock 2010 U	-0.00436	-0.00440	-0.00452
	(0.00796)	(0.00847)	(0.00863)
interaction Patent * TM stocks	0.00493***	0.00493***	0.00508**
	(0.00138)	(0.00127)	(0.00216)
Breadth TM U	0.00800	0.00807	
	(0.00493)	(0.0107)	
Breadth TM U squared		-1.36e-05	
		(0.00119)	
Sectoral Patent stock 2010 U	-0.0208**	-0.0207**	-0.0461***
	(0.0102)	(0.0102)	(0.0134)
Sectoral TM stock 2010 U	0.0585***	0.0585***	0.135***
	(0.0120)	(0.0124)	(0.0404)
Sectoral breadth TM U	0.243**	0.243***	
	(0.102)	(0.0941)	
US dummy	0.0566***	0.0566***	0.0520***
	(0.0105)	(0.0118)	(0.0133)
Service intensity U 2008-10			0.0607***
			(0.0217)
Constant	0.372*	0.371*	0.520***
	(0.214)	(0.203)	(0.111)
Sector dummies	YES	YES	YES
Pseudo R2	0.0731	0.0731	0.737
Observations	1,960	1,960	1,960

Table 8 – Using patent variables only: LAD quantile regression estimates for models testing Hypotheses 4 and 5 (median regression estimates reported).

Figure 4 – Estimated coefficients of Patent and trademark stock variables evaluated at different quantiles, confidence intervals included.



Dashed line represents OLS estimates with corresponding confidence intervals (dotted lines).

5.4 Results for High-tech, medium-tech and low-tech sectors

As a further analysis, we follow the strategy of Montresor and Vezzani (2015) and investigate whether the estimated effects differ across groups of sectors, namely high tech vs medium-high tech vs low tech sectors (including low tech and medium-low tech).

The results in Tables 9 clearly indicate that the validity of the hypothesis differs across groups of sectors. In fact, for low tech sectors, no significant effect is found for any of our variables of theoretical interest.

As for the hypotheses on market capabilities, a positive relation with net sales growth is found both for high tech and medium-high tech sectors, the only difference being that the sectoral effect of trademark stocks becomes negative for medium-high tech sectors.

As for the hypotheses on diversification, the relation between breadth and net sales growth is not significant, as it was in the overall sample. The results for Model 6 show how the service intensity is indeed positively related to net sales growth in high tech sectors. Instead, the variable is insignificant in medium-high tech sectors. What these results say is that the premium effect of providing services is mostly there for the most R&D intensive activities, notably manufacturing sectors where there is a strong focus on technology-based innovation.

HIGH TECH	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
Ln Net sales 2010	-0.0138	-0.0171	-0.0247	-0.0166	-0.0174	-0.0246
Ln R&D stock 2010	(0.0174) -0.0276* (0.0160)	(0.0187) -0.0411** (0.0166)	-0.0594*** (0.0176)	-0.0415** (0.0163)	(0.0185) -0.0410** (0.0165)	-0.0353** (0.0160)
Ln TM stock 2010 U		0.0247*** (0.00918)	-0.0583 (0.0459)	0.0256** (0.0113)	0.0245** (0.0124)	0.0230** (0.00965)
Interaction R&D * TM stocks			0.00643*			
Sectoral ln R&D 2010 HT	-0.814***	-0.444***	(0.00343) -0.435***	-0.455***	-0.444***	-0.401***
Sectoral TM stock U HT	(0.143)	(0.0981) 0.112***	(0.0985) 0.0897**	(0.0990) 0.115***	(0.101) 0.111***	(0.105) 0.107**
US dummy	0.0799***	(0.0416) 0.0508** (0.0222)	(0.0436) 0.0476** (0.0228)	(0.0415) 0.0505* (0.0265)	(0.0404) 0.0466* (0.0264)	(0.0422) 0.0517** (0.0224)
Breadth TM U	(0.0195)	(0.0232)	(0.0238)	(0.0283) -0.00403 (0.0117)	(0.0284) 0.00887 (0.0317)	(0.0254)
Breadth TM U squared					-0.00162 (0.00592)	
Service intensity U 2008-10						0.0787*
Constant	5.053*** (0.775)	2.923*** (0.479)	3.262*** (0.461)	2.980*** (0.487)	2.922*** (0.501)	(0.0432) 2.720*** (0.520)
Sector dummies	YES	YES	YES	YES	YES	YES
Pseudo R ² Observations	0.0750 683	0.0819 683	0.0843 683	0.0819 683	0.0821 683	0.0840 683

Table 9.1 – High-tech sectors only: LAD quantile regression estimates for models testing Hypotheses 1, 2 and 3.

MEDIUM HIGH TECH	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
Ln Net sales 2010	-0.0145 (0.0124)	-0.0267** (0.0104)	-0.0238** (0.0101)	-0.0265** (0.0104)	-0.0262** (0.0106)	-0.0201** (0.00940)
Ln R&D stock 2010	-0.00203	-0.0106	-0.0406**	-0.0102	-0.00964	-0.0147
Ln TM stock 2010 U	(0.0110)	0.0266***	-0.0622	0.0224***	0.0207**	0.0218***
Interaction R&D * TM stocks		(0.00704)	(0.0454) 0.00696**	(0.00831)	(0.00819)	(0.00693)
Sectoral ln R&D 2010 MT	0.0511***	0.0541***	(0.00352) 0.0529*** (0.0108)	0.0560***	0.0549***	0.0442***
Sectoral TM stock U MT	(0.00012)	-0.0285**	-0.0243*	-0.0243	-0.0231	-0.0243*
US dummy	0.0990***	0.0628***	0.0690***	0.0696***	0.0709***	0.0564***
Breadth TM U	(0.0166)	(0.0213)	(0.0214)	(0.0210) 0.00603	(0.0198) 0.0117	(0.0211)
Breadth TM U squared				(0.00666)	(0.0180) -0.00077 (0.00246)	
Service intensity U 2008-10						0.0639* (0.0365)
Constant	0.0338	0.334***	0.658***	0.310***	0.301***	0.337***
Sector dummies	YES	YES	YES	YES	YES	YES
Pseudo R ² Observations	0.0691 614	0.0868 614	0.0927 614	0.0873 614	0.0879 614	0.0888 614

Table 9.2 – Medium high-tech sectors only: LAD quantile regression estimates.

LOW TECH	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6
Ln Net sales 2010	0.00367	0.00399	0.00195	0.00602	0.00444	0.00615
Ln R&D stock 2010	(0.0140) -0.0132 (0.0173)	(0.0101) -0.0136 (0.0103)	(0.0135) 0.0126 (0.0235)	(0.0132) -0.0142 (0.0146)	(0.0114) -0.0132 (0.0149)	(0.0144) -0.0147 (0.0181)
Ln TM stock 2010 U	(0.0175)	-0.00099 (0.00620)	(0.0235) 0.0840 (0.0708)	-0.00458 (0.00793)	0.00160	-0.00277 (0.00646)
Breadth TM U		()	(0.00357 (0.00748)	-0.0200 (0.0215)	()
Sectoral ln R&D 2010 LT	-0.0128 (0.0534)	-0.0126 (0.0278)	-0.00541 (0.0441)	-0.00552 (0.134)	-0.00606 (0.143)	-0.0122 (0.0638)
Sectoral TM stock U LT		-0.00397 (0.0248)	-0.00793 (0.0328)	0.000760 (0.0928)	-0.00021 (0.0957)	0.00307 (0.0351)
Sectoral breadth TM U LT				-0.0215 (0.418)	-0.0131 (0.433)	
US dummy	0.0802*** (0.0295)	0.0809*** (0.0253)	0.0836*** (0.0286)	0.0813*** (0.0283)	0.0804*** (0.0279)	0.0742** (0.0291)
Interaction R&D * TM stocks			-0.00657 (0.00541)			
Breadth TM U squared					0.00355 (0.00332)	
Service intensity U 2008-10						0.0204 (0.0603)
Constant	0.239 (0.339)	0.251 (0.203)	-0.0708 (0.365)	0.208 (0.579)	0.223 (0.604)	0.202 (0.377)
Sector dummies	YES	YES	YES	YES	YES	YES
Pseudo R ² Observations	0.2437 236	0.2440 236	0.2503 236	0.2442 236	0.2474 236	0.2447 236

Table 9.3 – Low-tech sectors only: LAD quantile regression estimates.

6. Discussion and conclusions

This study has offered a novel and original analysis of the determinants of sales growth for top corporate R&D investors, exploiting the rich dataset on top R&D investors and their use of IPRs made possible by the joint efforts of EC JRC and OECD. This study contributes to an emerging literature analysing the economic value of intangibles for companies with an analysis of growth determinants, instead of market value antecedents.

Overall, the results indicate three key insights on the drivers of sales growth of top R&D investors. Firstly, companies that have filed more trademarks than the median firm do not enjoy a direct growth premium, but rather benefit from the positive effect that these market capabilities have on the relation between R&D investment and growth. This moderating effect overshadows the direct effect of trademark stocks on sales growth that we expected in the first place. What the results suggest is also that technological capabilities are high-demanding resources that are not directly linked to short-term changes in gains from market sales. Technological capabilities either measured by R&D or by patents, need to be further translated into actual market opportunities before realizing revenues in the marketplace. Companies with larger trademark portfolios appear to be better able to exploit their technological capabilities towards market growth.

Secondly, top R&D investors that have focused on service markets enjoy a clear growth premium. This is both a sectoral and a firm effect. In certain sectors, offering service is simply more valuable than in other sectors. In particular we find that the general results are driven by the effect of service intensity in high-tech sectors. Especially in these sectors, firms that own relatively more service marks than other firms in the same sector, enjoy a clear growth premium. These results confirm our expectations that top R&D investors, especially the ones focused mostly on delivering technology-based product innovations, benefit from designing service-centred offers as well. Earlier studies (Schmoch, 2003) have convincingly put forward the idea that trademark data can help in capturing service innovation. In this interpretation, the use of trademarks next to R&D and patents allows us to capture a broader notion of innovation, also including new services. We did not investigate the nature of the services: they may range from simple add-on services like repair solutions, to fullyfledged product-service solutions delivering high value to customers. Yet, these results are in line with recent literature stressing the benefits of organizational change towards a service-based logic, which ultimately is a more client-centred logic (Cusumano et al., 2015; Janssen, Castaldi and Alexiev, 2016). In doing so we also offer a novel measure of servitization, which has been measured so far in very different ways with no preferred indicator available yet (Gebauer, Fleisch and Friedli, 2005). This new measure can be widely applied within management studies.

Thirdly, using quantile regressions allowed us to gauge the differences in effects across the growth distributions. Indeed we found significant differences in the effects of some of our key variables of theoretical interest, reminding us that the baseline relations valid for median firms are hardly general laws applicable to firms experiencing variegated growth performances. We also found that our theoretical hypotheses are mostly confirmed in high-tech and medium-high tech sectors. Top R&D investors in low-tech sectors seem to follow very different dynamics and apparently require further theoretical efforts.

Our results offer insights from a managerial as well as a policy perspective. On the one hand, they indicate that both technological capabilities and market capabilities matter to improve the impact of firms' innovation activities on their growth performance. On the other hand the strategic management of innovation, as a key component of firm strategy (Hamel, 2000), also implies thinking both in terms of services as well as of delivering products in order to enhance the ability to meet market demand. Such dual approach is even more essential, also for high-tech manufacturing firms, as business value and consumer loyalty increasingly depends on the range of services offerings that come with the technology-based products. From a policy perspective, our study supports innovation policies that help companies to increase their service orientation; this is true also and maybe even more for companies from R&D-intensive manufacturing industries. Recent evidence on the experience of high-tech regions in Europe shows that several policy initiatives are already in place towards this end (Janssen and Castaldi, 2018). These regional policies could be inspirational for other regions.

Our results on the role of trademark breadth did not find any support for our hypothesis. With a different interpretation of breadth, namely as a measure of trademark value, also Sandner and Block (2011) find insignificant relation of trademark breadth and market value of companies. One problem with measures of breadth is that they depend very much on the branding strategies. Whenever companies opt for 'umbrella' type of brands, the related trademarks will tend to be 'broader', i.e. covering more classes, than in the cases of 'house of brands' strategies, when companies apply for new trademarks for single new products (or product families). Ideally, one should condition for these different types of strategies and they typically do not simply differ across sectors but rather across firms. With the available data, it was not possible to satisfactorily control for firm-level differences in branding strategies. This could partly explain the insignificant results. It is also the case that possibly better measures of diversification, based on Herfindhal indexes would have the same limitation even though exploiting the full information on the distribution of trademark applications over Nice classes. Castaldi and Giarratana (2018) explore a definition of branding strategies relying on both trademark breath and trademark intensity in a panel dataset of management consulting firms. Similar exercises for multi-sector samples of firms are trickier since a low number of trademarks could be a result of low sectoral trademark intensity, low firm trademark intensity or poor market capabilities. Nevertheless, further research should develop alternative indicators of diversification.

References

- Allegrezza, S. & Guarda-Rauchs, A. (1999). The determinants of trademarks deposits: An econometric investigation (A case study of the Benelux). *Economie Appliquée*, 52(2), 51-68.
- Akerlof, G. (1970). The market for "lemons": quality uncertainty and the market mechanism, Quarterly Journal of Economics, 84, 488-500.
- Appelt, S. (2009). Early Entry and Trademark Protection: An Empirical Examination of Barriers to Generic Entry, Paper presented at the DRUID Summer Conference 2009.
- Arora, A. and Nandkumar, A. (2012). Insecure advantage? Markets for technology and the value of resources for entrepreneurial ventures. Strategic Management Journal, 33(3), 231-251.
- Block, J. H., Fisch, C. & Sandner, P. G. (2014). Trademark families: Characteristics and market values. Journal of Brand Management, 21(2), 150-170.
- Block, J. H., Fisch, C. O., Hahn, A. & Sandner, P. G. (2015). Why do SMEs file trademarks? Insights from firms in innovative industries. Research Policy, 44(10), 1915-1930.
- Bottazzi, G. & Secchi, A. (2006). Explaining the distribution of firm growth rates. The RAND Journal of Economics, 37(2), 235-256.
- Breschi, S., Lissoni, F. & Malerba, F. (2003). Knowledge-relatedness in firm technological diversification. Research Policy, 32, 69-87.
- Castaldi, C. (2018), To trademark or not to trademark: the case of the creative and cultural industries, Research Policy, forthcoming.
- Castaldi, C. & Giarratana, M.S. (2018). Diversification, branding and performance of professional service firms, Journal of Service Research, forthcoming.
- Coad, A.. & Rao, R. (2008), Innovation and firm growth in high-tech sectors: A quantile regression approach. Research Policy, 37(4), 633-648.
- Corrado, C. A. & Hulten, C. R. (2010). How Do You Measure a" Technological Revolution"?. The American Economic Review, 100(2), 99-104.
- Cusumano, M. A., Kahl, S. J. & Suarez, F. F. (2015). Services, industry evolution, and the competitive strategies of product firms. Strategic Management Journal, 36(4), 559-575.
- Datta, A., Mukherjee, D. & Jessup, L. (2015). Understanding commercialization of technological innovation: taking stock and moving forward. R&D Management, 45(3), 215-249.
- Dernis H., Dosso M., Hervás F., Millot V., Squicciarini M. & Vezzani A. (2015). World Corporate Top R&D Investors: Innovation and IP bundles. A JRC and OECD common report. Luxembourg: Publications Office of the European Union.
- Dosso M., Vezzani A. (2017). Firm market valuation and intellectual property assets. JRC Working Papers on Corporate R&D and Innovation, No 07/2017, Joint Research Centre.
- Economides (1988). The Economics of Trademarks. Trademark Reporter, 78, 523-539.
- European Patent Office (EPO) and Office for the Harmonization in the Internal market (OHIM) (2013). Intellectual property rights intensive industries: contribution to economic performance and employment in the European Union, September 2013, Joint project between the EPO and the OHIM.
- European Commission (2013). The 2013 EU Industrial R&D Investment Scoreboard, Luxembourg: Publications Office of the European Union.
- European Commission (EC) (2015), Innovation Union Scoreboard 2015.
- Flikkema, M.J., de Man, A.P. & C. Castaldi (2014). Are Trademark Counts a Valid Indicator of Innovation? Results of an In-depth Study of New Benelux Trademarks Filed by SMEs. Industry and Innovation, 21(4), 310-331.
- Flikkema, M., Castaldi, C., De Man, A. P. & Seip, M. (2015). Explaining the Trademark-Innovation Linkage: the Role of Patents and Trademark Filing Strategies, DRUID Society conference 2015, http://druid8.sit.aau.dk/acc_papers/nv9r3f3sl6p5e0431rj6sxfblv4g.pdf
- Gao, G. & Hitt, L. M. (2012). Information technology and trademarks: Implications for product variety. Management Science, 58(6), 1211-1226.
- Gebauer, H., Fleisch, E. & Friedli, T., (2005). Overcoming the service paradox in manufacturing companies. European Management Journal, 23(1), 14–26.
- Giarratana, M. S. & Fosfuri, A. (2007). Product strategies and survival in Schumpeterian environments: Evidence from the US security software industry. Organization Studies, 28(6), 909-929.
- Götsch, M. & Hipp, C. (2012). Measurement of innovation activities in the knowledge-intensive services industry: a trademark approach. The Service Industries Journal, 32(13), pp. 2167-2184.

- Graham, S. J., Hancock, G., Marco, A. C. & Myers, A. F. (2013). The USPTO trademark case files dataset: Descriptions, lessons, and insights. Journal of Economics & Management Strategy, 22(4), 669-705.
- Graham, S. J., Marco, A. C. & Myers, A. F. (2015). Monetizing Marks: Insights from the USPTO Trademark Assignment Dataset. Available at SSRN 2430962.
- Greenhalgh, C., & Rogers, M. (2010). Innovation, intellectual property, and economic growth. Princeton University Press.
- Greenhalgh, C. & Rogers, M. (2012). Trade Marks and Performance in Services and Manufacturing Firms: Evidence of Schumpeterian Competition through Innovation. The Australian Economic Review, 45(1), 50-76.
- Greenhalgh, C., Rogers, M., Schautschick, P. & Sena, V. (2011). Trade mark incentives. Intellectual Property Office Research Paper, (2011/1).
- Haans, R. F., Pieters, C., & He, Z. L. (2016). Thinking about U: Theorizing and testing U-and inverted U-shaped relationships in strategy research. Strategic Management Journal, 37(7), 1177-1195.
- Hall, B.H. & Oriani, R. (2006). Does the market value R&D investment by European firms? Evidence from a panel of manufacturing firms in France, Germany, and Italy. International Journal of Industrial Organization, 24(5), 971–993.
- Hamel, G. (2000). Leading the revolution. Boston, MA: Harvard Business School Press.
- Hipp, C. & Grupp, H. (2005). Innovation in the service sector: The demand for service-specific innovation measurement concepts and typologies, Research Policy, 34, 517-535.
- Huang, P., Ceccagnoli, M., Forman, C. & Wu, D. J. (2013). Appropriability mechanisms and the platform partnership decision: Evidence from enterprise software. Management Science, 59(1), 102-121.
- Hurmelinna-Laukkanen, P., & Puumalainen, K. (2007). Nature and dynamics of appropriability: strategies for appropriating returns on innovation. R&D Management, 37(2), 95-112.
- Janssen, M.J. & Castaldi, C. (2018). Services, innovation, capabilities and policy: Towards a synthesis and beyond, Science and Public Policy, forthcoming.
- Janssen, M. J., Castaldi, C. & Alexiev, A. (2016). Dynamic capabilities for service innovation: conceptualization and measurement. R&D Management, 46(4), 797-811.
- Janssen, M. J. & den Hertog, P. (2016). Developing Service-Based Business Models: Which Innovation Capability for Which Innovation Dimension?. In Service Innovation (pp. 97-128). Springer Japan.
- Keller, K.L. (1993). Conceptualizing, Measuring, and Managing Customer-Based Brand Equity. Journal of Marketing, 57, 1–22.
- Keller, K.L. & Lehmann, D.R. (2006). Brands and Branding: Research Findings and Future Priorities. Marketing Science, 25(6): 740–759.
- Llerena, P. & Millot, V. (2013). Are Trade Marks and Patents Complementary or Substitute Protections for Innovation. BETA WP No. 2013-01, UDS, Strasbourg.
- Milgrom, P. & Roberts, J. (1986). Price and advertising signals of product quality. The Journal of Political Economy, 796-821.
- Millot, V. (2012). Trademark strategies and innovative activities. Ph.D. thesis, Université de Strasbourg.
- Millot, V. (2009). Trademarks as an indicator of product and marketing innovations (No. 2009/6). OECD Publishing.
- Malmberg, C. (2005). Trademark statistics as innovation indicator? A micro study. CIRCLE, WP 2005/17.
- Mendonça S., Pereira, T.S. & Godinho, M.M. (2004). Trademarks as an indicator of innovation and industrial change. Research Policy, 33, 1385-1404.
- Mendonça, S. (2014). National adaptive advantages: Soft innovation and marketing capabilities in periods of crisis and change. In A.A.C. Teixeira, E. Silva, R. Mamede (Eds), Structural Change, Competitiveness and Industrial Policy: Painful Lessons from the European Periphery, Routledge, 128-45.
- Montresor, S. & Vezzani, A. (2015). The production function of top R&D investors: Accounting for size and sector heterogeneity with quantile estimations. Research Policy, 44(2), 381-393.
- Nam, Y. & Barnett, G. A. (2011). Globalization of technology: Network analysis of global patents and trademarks. Technological Forecasting and Social Change, 78(8), 1471-1485.
- OECD (2015). OECD Science, Technology and Industry Scoreboard 2015: Innovation for growth and society, OECD Publishing, Paris.

- Palich, L. E., Cardinal, L. B., & Miller, C. C. (2000). Curvilinearity in the diversification-performance linkage: an examination of over three decades of research. Strategic Management Journal, 21(2), 155-174.
- Ramello, G.B. (2006). What's in a sign? Trademark law and economic theory. Journal of Economic Surveys, 20(4), 547-565.
- Ramello, G. B. & Silva, F. (2006). Appropriating signs and meaning: the elusive economics of trademark. Industrial and Corporate Change, 15(6), 937-963.
- Reitzig, M. (2004). Strategic Management of Intellectual Property. Sloan Management Review Spring 2004, 35-40.
- Sandner, P.G. & Block, J. (2011). The market value of R&D, patents, and trademarks. Research Policy, 40 (7), 969-985.
- Schautschick, P., & Greenhalgh, C. (2016). Empirical studies of trade marks-the existing economic literature. Economics of Innovation and New Technology, 25(4), 358-390.
- Schmoch, U. (2003). Service marks as novel innovation indicator. Research Evaluation, 12, 149-156.
- Schmoch, U. & Gauch, S. (2009). Service marks as indicators for innovation in knowledge-based services. Research Evaluation, 18(4), 323-335.
- Scholec M. & Verspagen B (2012). The Voyage of the Beagle into innovation: explorations on heterogeneity, selection, and sectors. Industrial and Corporate Change 21(5), 1221-1253.
- Teece, D.J. (1982). Towards an economic theory of the multiproduct firm, Journal of Economic Behavior & Organization, 3, 39-63.
- Teece, D. J., Rumelt, R., Dosi, G. & Winter, S. (1994). Understanding corporate coherence: Theory and evidence. Journal of Economic Behavior & Organization, 23(1), 1-30.
- Vandermerwe, S. & Rada, J. (1989). Servitization of business: adding value by adding services. European Management Journal, 6(4), 314-324.
- Vomberg, A., Homburg, C. & Bornemann, T. (2015). Talented people and strong brands: The contribution of human capital and brand equity to firm value, Strategic Management Journal, 36 (13), 2122-31.
- WIPO, World Intellectual Property Organization (2004), Intellectual Property Handbook, http://www.wipo.int/export/sites/www/freepublications/en/intproperty/489/wipo_pub_489 .pdf
- WIPO (2013a). Nice classification: The International Classification of Goods and Services for the Purposes of the Registration of Marks.
- WIPO (2013b), World Intellectual Property Report 2013 Brands Reputation and Image in the Global Marketplace.
- Yaffee, R. A. (2002). Robust regression analysis: some popular statistical package options. ITS Statistics, Social Science and Mapping Group, New York State University.
- Zhou, H. Sandner, P.G., Martinelli, S.L. & Block J. H. (2016). Patents, trademarks, and their complementarity in venture capital funding, Technovation, 47, 14-22.

JRC Mission

As the science and knowledge service of the European Commission, the Joint Research Centre's mission is to support EU policies with independent evidence throughout the whole policy cycle.



EU Science Hub ec.europa.eu/jrc

- 9 @EU_ScienceHub
- **f** EU Science Hub Joint Research Centre
- in Joint Research Centre
- EU Science Hub